Author's response to reviews

Title: Is There a Role for Expectation Maximization Imputation in Addressing Missing Data in Research Using WOMAC Questionnaire? Comparison to the Standard Mean Approach and a Tutorial

Authors:

Hassan MK Ghomrawi (hag2008@med.cornell.edu)
Lisa A Mandl (Mandll@hss.edu)
John Rutledge (jrr2002@med.cornell.edu)
Michael M Alexiades (AlexiadesM@HSS.EDU)
Madhu Mazumdar (mam2073@med.cornell.edu)

Version: 4 Date: 20 May 2011

Author's response to reviews:

Dear Editor,

Thank you for accepting our manuscript for publication in your Journal. We have addressed in this revised version the formatting comments needed before the manuscript is deemed ready for publication.

Regards,
Hassan Ghomrawi